Author's response to reviews

Title: Protocol for a Systematic Review on the association between chronic stress during the life course and telomere length

Authors:

Jacklyn Quinlan (jacki.quinlan@gmail.com)
Mai Thanh Tu (mai.thanh.tu@umontreal.ca)
Etienne V Langlois (etienne.langlois@umontreal.ca)
Mohit Kapoor (mohit.kapoor.chum@ssss.gouv.qc.ca)
Daniela Ziegler (daniela.ziegler.chum@ssss.gouv.qc.ca)
Hassan Fahmi (h.fahmi@umontreal.ca)
Maria-Victoria Zunzunegui (maria.victoria.zunzunegui@umontreal.ca)

Version: 4 Date: 14 April 2014

Author's response to reviews: see over
April 14, 2014

RE: Revisions to submission  
MS: 9296322881182938  
Protocol for a Systematic Review on the association between chronic stress during the life course and telomere length  
Jacklyn Quinlan, Mai Thanh Tu, Etienne V Langlois, Mohit Kapoor, Daniela Ziegler, Hassan Fahmi and Maria-Victoria Zunzunegui

Dear editors of Systematic Reviews,

We are thankful for the comments we received on our submission to Systematic Reviews entitled “Protocol for a Systematic Review on the association between chronic stress during the life course and telomere length” and have now registered with PROSPERO.

We appreciate your consideration. Sincerely,

Jacklyn Quinlan  
Post-doctoral fellow  
School of Public Health  
University of Montreal

Maria-Victoria Zunzunegui  
Professeur  
School of Public Health  
University of Montreal
Title: Protocol for a Systematic Review on the association between chronic stress during the life course and telomere length

Version: 3

Date: 3 April 2014

Reviewer: Alexander Tsertsvadze

Response to reviewer’s report:

1. Please, register the protocol with the PROSPERO. The protocol is now registered with the PROSPERO (registration number: CRD42014009274)

2. The following sentence “This protocol will be registered with the PROSPERO database (registration number will be included once the review of the protocol has been finalised)” is repeated twice in Methods (page 8) and Discussion (page 14). Please, delete it on page 14.

We removed the sentence from page 14 and now include the following sentence on page 8:
This protocol is registered with the PROSPERO database (registration number: CRD42014009274).